Author’s response to reviews

Title: A Bayesian Decision Support Sequential Model for Severity of Illness Predictors and Intensive Care Admissions in Pneumonia Short/ Running Title: Bayesian Pneumonia Decision

Authors:

Amado Baez (aabaezmd@gmail.com)
Laila Cochon (lailacochn@hotmail.com)
Jose Nicolas (nicolas@ub.edu)

Version: 2 Date: 16 Dec 2019

Author’s response to reviews:

Thank you for the opportunity to revise our manuscript. Below please find decision letter comments and our answers.

1. Please make sure the abstract contains only the following headings: Background, Methods, Results, Conclusions.

Yes, edited

2. If abbreviations are used in the text they should be defined in the text at first use, and a list of abbreviations should be provided in the Declarations.

Checked and done

3. Please note, the "Availability of Data and Materials" declaration refers to the raw data used in your study.

We strongly encourage all authors to share their raw data, either by providing it in a supplementary file or depositing it in a public repository and providing the details on how to access it in this section.

If you do not wish to share your data, please clearly state this in this section along with a justification. For more information and a list of suitable availability statements, please see our Submission guidelines: https://www.biomedcentral.com/getpublished/editorial-policies#availability+of+data+and+materials

The manuscript reports a mathematical method and no data set used.
4. Please include the full name of the ethics committee (and the institute to which it belongs to) that approved the study and the committee’s reference number if appropriate.

Full information on ethics and letter number enclosed.

5. Consent for publication refers to consent for the publication of identifying images or other personal or clinical details of participants that compromise anonymity. If your manuscript uses identifying images or other personal/clinical details, please include a statement of consent to publish from the patient, or in the case of minors, the patients’ guardians. If this is not applicable to your manuscript please state “Not Applicable” in this section.

NA stated

6. Please provide a COMPLETE Declarations section before the references.

Please note that we require ALL of the sections to be present (unless stated optional), as well as a “Declarations” heading.

Complete stated

Please find instructions for completing the declarations in the submission guidelines:

The Declarations sub-sections are:

- Ethics approval and consent to participate
- Consent for publication
- Availability of data and materials
- Funding
- Competing Interests
- Authors’ contributions
- Acknowledgements

If any of the sections are not relevant to your manuscript, please state "Not applicable" in the section. DONE
7. Please include a statement in the Authors' contributions section to the effect that all authors have read and approved the manuscript, and ensure that this is the case. DONE

8. Please remove the conflict of interest and funding statements from the title page. DONE

9. At this stage, please upload your manuscript as a single, clean version that does not contain any tracked changes, comments, highlights, strikethroughs or text in different colours. All relevant tables/figures/additional files should also be clean versions. Figures (and additional files) should remain uploaded as separate files.